TPOL: stable operation for the most part, continuous coverage over Christmas

0l

==l =l
File
History plat Set timea scala
lepton current [ma) tirme window labals ticks
Folarization(ally [3]
LPOL polarisation [% £ 10.0 h « 1l0.0h -, 10h
a0 N 50 h «w 5.0h w  30.0min
N 2.0h 4 Z2.0h <> 10,0 min
N 1.0 h « Ll0Oh s S0 min
50 Select variables Change Y-axis
short list J Polarization(all) [%]
LPOL/TROL ratic - Hide axis
2o Off Rate _ Isericale
On Rate "
Folarization(all) |¢00
Palarizationizcllidin =
Folarizationinon —co |:| 00.0
Yert, emittance colli
Smaoothing
[ D "
| 0.0

update display |

Data di

Choose data set Change Y-axis

bunch polarisa -
’ _'J linear lightpo! diode [5]

B0

&0

40

20

0.5

LPOLFTPOL ratio 5 n
coll, LPOL Smin avy
coll, bunch pol, 10n
coll. bunch pol, Imi

_ Hide axis

_ UserScale

coll, bunch pol, Smi
lepton bunch curren

]
- 0 i50.0
update display |

online display from fill yesterday morning




communication problem with HERA:
problem with netmex server on TPOL side: fixed

need to improve the monitoring

problem to update the ORACLE database
post-mortem fix exists (so all data will be entered into the database)
changes to the computer infrastructure means that automatic
update at the moment are disabled: working on solution



LPOL calorimeter installed in tunnel last week:
re-commissioning without problems, stable operations:
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LPOL Status Information
Ll eaon |
POL.

OK, measurement in progress
% Compton Calo PMT1
Calo PMTZ |
Calo PMT3
Calo PMT4
Calo Sum
Energy Res.

17.02 %

1.46 %
-0.96

-0.19 ns
20.99

0.04 mm
0.03 mm

-0.40 mm
1.60 mm
0.00 %

LPO

Rel. Lumi:




Hardware continues to be commissioned:
during last 2 accesses temperature stabilization installed and commissioned

Temeprature measured at the input mirror:
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System works basically, further funing needed.



Intense discussions and simulations ongoing

optimise the geometry,
understand the resolution

potential problem: low light yield of Cherencov fibers
result in poor resolution

Goal: about 20-25%/sqrt(E) resolution is adequate for LPOL cavity

Finish design ASAP, start production



New feature implemented in TPOL online monitor: trend display

goal: provide fast "trend" information to see the direction of

change of the polarisation.

accessible after enabling expert menu under

-> frames -> Polarimeter trend

feedback very welcome,
whether this is a useful
feature, whether it should be
improved, ....
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Chi==2/MDF: quality of risatime fit

Assumption: tau_st=40 minutes

Slopa: polarisation change per minute from linear fit
Ficurrant) current polarisation extrapolated from linear fit

Finfinity) equilibrium polarisation from risetime fit

Caution: the risetime fit has not bean testad well




